\documentclass[a4paper]{article}

\usepackage[utf8]{inputenc}
\usepackage[ngerman]{babel}
%\usepackage{times}
%\usepackage{multicol}
\usepackage{amssymb} % for $\square$
\usepackage{palatino}

\oddsidemargin -20pt
\evensidemargin -20pt
\topmargin -75pt
\advance\textwidth5cm
\advance\textheight6cm

%###############
% New Commands #
%###############

\newcommand \rack {{\LARGE $\square$}}

\newcommand{\ebf}[6] {
  \begin{tabular}{rccccccl}
    \rack & \rack & \rack & \rack & \rack & \rack \\%
    #1 & #2 & #3 & #4 & #5 & #6 % \tabularnewline
  \end{tabular}
}
\newcommand{\ebftina}[6] {
  \begin{tabular}{rccccccl}
    &\rack & \rack & \rack & \rack & \rack & \rack \\
    &#1 & #2 & #3 & #4 & #5 & #6  % \tabularnewline
  \end{tabular}
}
\newcommand{\ebffuenf}[5] {
  \begin{tabular}{rcccccl}
    &\rack & \rack & \rack & \rack & \rack \\
    &#1 & #2 & #3 & #4 & #5  % \tabularnewline
  \end{tabular}
}

\newcommand{\ebfleistung}[6] {
  \begin{tabular}{rcccccl}
    &\rack & \rack & \rack & \rack & \rack & \rack \\
    &#1 & #2 & #3 & #4 & #5 & #6 % \tabularnewline
  \end{tabular}
}

\newcommand{\ebfleistunggross}[8] {
  \begin{tabular}{rcccccccl}
    &\rack & \rack & \rack & \rack & \rack & \rack & \rack & \rack\\
    &#1 & #2 & #3 & #4 & #5  & #6 &  #7 & #8 % \tabularnewline
  \end{tabular}
}

\newcommand{\manswers} {
  \hspace*{0.4cm}(Hier sind mehrere Antworten erlaubt.)}

\newcommand{\citem}[1] {
  \hspace*{0.5cm}
  \raisebox{-.25em}[.25em]{\rack}
  %\hspace*{0.1cm}
  {#1}\ \ \phantom{\dotfill}\\[.4em]
}

\newcommand{\citemtable}[1] {
  \hspace*{0.5cm}
  \raisebox{-.25em}[.25em]{\rack}
  %\hspace*{0.1cm}
  {#1}\ \ \phantom{\dotfill}
}

\newcommand{\citemdots}[1] {
  \hspace*{0.5cm}
  \raisebox{-.25em}[.25em]{\rack}
  %\hspace*{0.1cm}
  {#1}\ \ \dotfill \\[.4em]
}

\newcommand{\citemdotstable}[1] {
  \hspace*{0.5cm}
  \raisebox{-.25em}[.25em]{\rack}
  %\hspace*{0.1cm}
  {#1}\ \ \dotfill
}

\newcommand{\pitem}[1] {
\\
\underline{\hspace{1cm}}\%\hspace*{0.2cm} {#1}\\
}

\newcommand{\yn}[2]{
\hspace*{0.25cm} \rack \hspace*{0.1cm} {#1}\hspace*{0.25cm}\rack \hspace*{0.1cm} {#2}
}

\newcommand{\freetext}{\fbox{\rule[8cm]{.99\linewidth}{0cm}}}

\newcommand{\freetextsize}[1]{\fbox{\rule[#1]{.99\linewidth}{0cm}}}

%###############
% End Commands #
%###############


\begin{document}

\pagestyle{empty}

\begin{center}
{\bf\huge Course evaluation form \\}
%###########################################
%############ Fill me in! ##################
{\bf\Large Summer term 2011}
%############ Fill me in! ##################
%###########################################
\end{center}

\section*{Note:}

% Please fill in this form for each course you are attending. If there are several lecturers for the same course, please fill in one form per lecturer. You can submit the forms either to the lecturer or to the study advisor (Ms. Martina Welte, building 051 room 02-029) or the program co-ordinator (Ms. Ursula Epe, building 101 room 02-013a).
See the explanation in the forum for the course, under \emph{Official $\to$
Submit your evaluation form}.

\section{General}

{\bf Course name?}\ \ Efficient Route Planning, SS 2011\ \dotfill \\[.1cm]
{\bf Lecturer's name?}\ \ Prof.~Dr.~Hannah Bast\ \dotfill \\[.25cm]
\rack  \hspace*{0.1cm} {\bf \glqq I would like to nominate this lecturer for the teaching award. \grqq\\
My reason:}\\
\freetextsize{1.5cm} \\[.25cm]
{\bf Your semester?}\ \ \dotfill\\[.1cm]
{\bf Your study program:} \\
\begin{minipage}{0.48\textwidth}
\citemtable{Bachelor Informatik} \\
\citemtable{Bachelor ESE} \\
\citemtable{Master (Applied) Computer Science}
\end{minipage}
\begin{minipage}{0.5\textwidth}
 \citemtable{Staatsexamen} \\
 \citemdotstable{Diploma minor, major is:} \\
 \citemdotstable{other (i.e. minor course)}
\end{minipage}

\section{Questions about the course}

\begin{enumerate}

\item How would you rate the lecturer's over-all performance ? (preparation, motivation, explanation...)\\
very bad \ebffuenf {-2}{-1}{0}{1}{2}  very good

\item In your opinion: Is the content of the course well structured?\\
not at all \ebffuenf {-2}{-1}{0}{1}{2}  very well

\item How valuable is the accompanying material (slides, handout, recordings)?\\
not helpful \ebffuenf {-2}{-1}{0}{1}{2}  very helpful

\item How do you feel about this lecture?\\
not challenged \ebffuenf {-2}{-1}{0}{1}{2}  overstrained

\item How do you evaluate the quantity of material taught in this course?\\
not enough \ebffuenf {-2}{-1}{0}{1}{2}  too much

\item How many hours a week do you invest in this course (including lecture, tutorials, preparation, exercise sheets...)? \\
  \ebfleistunggross{0}{1-4}{5-8}{9-12}{13-16}{17-20}{21-24}{$>$25}

\newpage

\item What do you like about this course?

  \freetextsize{3cm}

\item What do you not like so much about this course?

  \freetextsize{3cm}

\item Any suggestions how to improve this course?

  \freetextsize{3cm}

\end{enumerate}

\section{Questions about tutorials (if available)}

\begin{enumerate}

\item In your opinion: How difficult are the exercises?\\
not difficult \ebffuenf {-2}{-1}{0}{1}{2}  very difficult

\item Are the tutorials a good addition to the lecture?\\
not at all \ebffuenf {-2}{-1}{0}{1}{2}  very good

\item How well does the tutor explain and answer questions?\\
not well \ebffuenf {-2}{-1}{0}{1}{2}  very well

\item In your opinion: Is the group size of the tutorial okay?\\
too small \ebffuenf {-2}{-1}{0}{1}{2}  too big

\item Your feedback concerning the exercises and the tutorial:

  \freetextsize{4cm}

\end{enumerate}

\end{document}

%%% Local Variables:
%%% mode: latex
%%% TeX-master: "kurzfragebogen"
%%% End: 
